Author's response to reviews

Title: Virtual Patient Simulation: what do students make of it? A focus group study.

Authors:

Mihaela Botezatu (mihaela.botezatu@ki.se)
Håkan Hult (hakan.hult@ki.se)
Uno G Fors (uno.fors@ki.se)

Version: 3 Date: 3 December 2010

Author's response to reviews:

Summary of changes

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name'(e.g. Joe F Bloggs). Currently they do not match.

-Changed on the title page, from Uno Fors to Uno G Fors. I did not find any other places where a change would have been in order.

Table 1: Please ensure that Table 1 is cited in the main manuscript text, rather than as just placement text.

-Text added at the end of the first paragraph in Results.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

-Removed.

Please do not number your headings or sub-headings.

-Removed where appropriate.

References: please provide the full citation details for reference 14. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references.

-Updated.

Tables- please remove the visible vertical lines from your tables.
Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements.

Text: Please ensure that all text is displayed in black throughout the manuscript.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.